Author's response to reviews

Title: Therapeutic and immunomodulatory activities of short-course treatment of murine visceral leishmaniasis with KALSOMETM10, a new liposomal amphotericin B

Authors:

Md Asad (asadseraphic.nii@gmail.com)
Pradyot Bhattacharya (p_pradyot@yahoo.com)
Nahid Ali (nali@iicb.res.in)
Antara Banerjee (banerjee.antara77@gmail.com)

Version: 5
Date: 20 February 2015

Author's response to reviews: see over
Dear Sir,

Please find attached our revised manuscript entitled “Therapeutic and immunomodulatory activities of a short-course treatment of murine visceral leishmaniasis with KALSOME™10, a new liposomal amphotericin B” by Asad et al. for consideration of publication in your esteemed journal. We have incorporated all the necessary corrections according to the journal’s requirement as suggested by the editor. We are highly thankful to the editorial board as well as the reviewers for critically reviewing our manuscript and helping us in improving it.

Thanking you.

Yours sincerely,

Nahid Ali, PhD